
\documentclass{amsart}
\newtheorem{theorem}{Theorem}[section]
\usepackage{tikz}
\usetikzlibrary{calc}
%\newtheorem{thHK}[theorem]{Theorem HK}

\usepackage{latexsym}
\usepackage{amssymb,amsmath,amsfonts}
%\usepackage[utf8x]{inputenc}
\usepackage[colorlinks=true]{hyperref}
\usepackage{graphicx}


%\newtheorem{thCM}[theorem]{Theorem CM}
%\newtheorem{theorC}[theorem]{Theorem C}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{cor}[theorem]{Corollary}
\newtheorem{prop}[theorem]{Proposition}
%\newtheorem{corCH}[theorem]{Corollary CH}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{Assumption}[theorem]{Assumption}
\newtheorem{construction}[theorem]{Construction}
%\newtheorem{note}[theorem]{}
\newtheorem{example}[theorem]{Example}
%\newtheorem{xca}[theorem]{Exercise}`

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\theoremstyle{proposition}
\newtheorem{proposition}[theorem]{Proposition}
\theoremstyle{corollary}
\newtheorem{corollary}[theorem]{Corollary}
\newcommand{\QED}{\hfill$\Box$}

\usetikzlibrary{positioning}
\usetikzlibrary{petri}
\usepackage{tikz}
\usetikzlibrary{shapes.arrows,arrows}

%\def\subjclassname{{\rm 2000} {\it Mathematics Subject Classification}}


\newtheorem*{result1}{{\bf Theorem S}}
\newtheorem*{result2}{{\bf Theorem I}}
\newtheorem*{result3}{{\bf Lemma MJ}}
\newtheorem*{result5}{{\bf Theorem B}}



\numberwithin{equation}{section}
\global\parskip 6pt \oddsidemargin
0.1cm \evensidemargin 0.4cm \headheight 0.1cm \topmargin -2.0cm
\textwidth 17.10cm \textheight 24.0cm \voffset 1.50cm \topskip 1.0cm

\begin{document}
\begin{thebibliography}{111}
\bibitem{Isra}{\rm I. Kleiner,} History of Group theory, {\it History of Abstract Algebra}, Birkhauser Boston, 17-39, (2007).
\bibitem{Jian}{\rm J. Zhang, F. Xiong and J. Kang}, The application of Group theory in communication operation pipeline system, {\it Mathematical problems in Engineering}, (2018).
\bibitem{Laan}{\rm J. Laane and E. J. Ocola,} Application of symmetry and group theory for the investigation of molecular vibrations, {\it Acta Applicandae Mathematicae}, {\bf 118}(1), 3-24, (2012).
\bibitem{Edwa}{\rm E. A. Rietman, R. L. Karp and J. A. Tuszynski,} Review and application of group theory to molecular system biology, {\it Theoretical Biology and medical modeling}, {\bf 8}(21), (2011).
\bibitem{Osbo}{\rm H. Osborn,} Symmetry relationships between Crystal Structures: application of crystallographic group theory in crystal chemistry, {\it Contemporary Physics}, {\bf 6}(1), 97-98, (2015).
\bibitem{Cal} {\rm A. Cayley,} Desiderata and suggestions: The theory of groups: graphical representation, {\it American Journal of Mathematics}, {\bf 1} (2) (1878) 403-405.
\bibitem{Kand}{\rm W. B. Vasantha Kandasamy and F. Samarandache}, Groups as Graphs, {\it Editura cuart and authors}, (2009).
\bibitem{Paul}{\rm P. H. Zieschang,} Cayley graph of finite groups, {\it Journal of Algebra}, {\bf 118}, 447 - 454 (1988).
\bibitem{Cam}{\rm P. J. Cameron and S. Ghosh,} The power graph of finite groups, {\it Discrete Mathematics,} {\bf 311}, 1220 -1222, (2011).
\bibitem{Reh}{\rm S. U. Rehman, A. Q. Baig, M. Imran and Z. U. Khan}, Order divisor graphs of finite groups, {\it An. St. Ovidus Constanta}, {\bf 26}(3), 29-40, (2018).
\bibitem{Wil}{\rm J. S. Williams,} Prime graph components of finite groups, {\it Journal of Algebra}, {\bf 69}(2), 487-513, (1981).
\bibitem{Xwa}{\rm X. L. Ma, H. Q. Wei and G. Zhong}, The cyclic graph of a finite groups, {\it Algebra}, (2013).
\bibitem{Tolu}{\rm A. Erfanian and B. Tolue,} Conjugate graphs of finite groups, {\it Discrete Mathematics, Algorithm and Applications}, {\bf 4}(2), (2012).
\bibitem{Akba}{\rm B. Akbari,} Hall graph of a finite group, {\it Note Mat.}, {\bf 39}(2), 25-37, (2019).
\bibitem{Andr}{\rm A. Lucchini,} The independence graph of a finite group, {\it Monatsheft Fur Mathematik}, {\bf 193}, 845-856, (2020).
\bibitem{Gore}{\rm D. Gorenstein,} Finite Groups, {\it Harper \& Row}, New York, 1968.
\bibitem{Robi}{\rm D. J. S. Robinson,} A course in the theory of Groups, {\it 2nd edition, Springer-Verlag, New York}, (1996). 
\bibitem{Dr} {\rm S. D. David and M. F. Richard}, Abstract algebra, 3rd Edition, {\it John Wiley and Son Inc.}, (2004).
\bibitem{Gb} {\rm C. Godsil and G. Boyle,} Algebraic graph theory, 5th edition, {\it Springer, Boston New York,} (2001).
\bibitem{Anj} {\rm A. Gupta,} Discrete mathematics, {\it S.K. Kataria \& Sons,} 2008, 258-310.
\bibitem{Cam1}{\rm P. J. Cameron,} Notes on finite group theory,{\it www.maths.qmul.ac.uk} (2013).
\bibitem{Rose}{\rm H. E. Rose,} A Course on Finite Groups; {\it Springer Science \& Business Media},(2009).
\bibitem{Clem}{\rm A. E. Clement, S. Majewicz and M. Zyman,} Introduction to Nilpotent Groups; {\it The Theory of Nilpotent Group. Birkhauser, Cham}, (2017).
\bibitem{Woj}{\rm W. A. Trybulec,} Commutator and Center of a Group, {\it Formalized Mathematics, Universite Catholique de Louvain}, {\bf 2}(4)(1991). 
\bibitem{Robe}{\rm R. M. Guralnick,} Commutators and Commutator Subgroups; {\it Advances in Mathematics}, {\bf 45}, 319-330 (1982).
\bibitem{Peo}{\rm people.math.binghamton.edu} (Accessed on 2nd October, 2020).
\bibitem{Note}{\rm www.math.columbia.edu} (Accessed 6th November, 2020).
\bibitem{Keit5}{\rm K. Conrad} Simplicity of $A_n$; {\it kconrad.math.uconn.edu} (Accessed on 7th November, 2020).
\bibitem{Step}{\rm S. R. Cavior,} The Subgroups of the Dihedral groups; {\it Mathematics Magazine}, {\bf 48}, 107, (1975).
\bibitem{Mari}{\rm M. Tarnauceanu,} A characterization of the quaternion group,{\it An. St. Univ. Ovidius Constanta}, {\bf 21}(1), 209-2014, (2013).

\section{Figure Legends}
\noindent {Figure 1:} Subgroups graph of $\mathbb{Z}_6 $; shows the relationships between subgroups of $\mathbb{Z}_6 $, integer group under addition modulo 6.\\

\noindent {Figure 2:} Subgroups graph of $\mathbb{Q}_8 $; shows the relationships between the subgroups of quaternion group.\\

\noindent These graphs clearly depict the Lagrange's theorem \cite{Rose} that if $H$ is a subgroup of a finite group $G$. Then the order of $H$ divides the order of $G$. 


















\end{thebibliography}

\vspace*{2cm}
\begin{center}
{\bf\large SUBGROUP GRAPHS OF FINITE GROUPS}
 \\[0.5cm]
  
{O. Ejima$^1$, A. I. Garba$^2$ and K. O. Aremu$^3$   \footnote{$^1$Corresponding Author :- unusoj1@yahoo.com} \\[2mm]
Department of Mathematics, Usmanu Danfodiyo University, Sokoto, Nigeria.\\
{\tt E-mail: unusoj1@yahoo.com}\\
{\tt E-mail: garba.isa@udusok.edu.ng}\\
{\tt E-mail: aremu.kazeem@udusok.edu.ng}} \\[2mm]
\end{center}%
\vspace*{0.5cm}
%

\title[SUBGROUP GRAPHS OF FINITE GROUPS]
{}
\address{$^{1,2,3}$ Department of Mathematics, Usmanu Danfodiyo University, Sokoto, Nigeria.}
\email{$^1$unusoj1@yahoo.com }
\email{$^{2}$garba.isa@udusok.edu.ng}
\email{$^{3}$aremu.kazeem@udusok.edu.ng}
\keywords{ Subgroup graphs; finite groups; commutator subgroup; graph of groups; graph of finite groups.\\
{\rm 2000} {\it Mathematics Subject Classification}: 05C25; 05C50.}
\footnote{Corresponding Author's E-mail:- unusoj1@yahoo.com}

\begin{abstract}
 Let $G$ be a finite group with the set of subgroups of $G$ denoted by $S(G)$, then the subgroup graphs of $G$ denoted by $\Gamma(G)$ is a graph which set of vertices is $S(G)$ such that two vertices $H, K \in S(G)$ $(H \neq K)$ are adjacent if either $H$ is a subgroup of $K$ or $K$ is a subgroup of $H$. In this paper, we introduce the Subgroup graphs $\Gamma$ associated with $G$. We investigate some algebraic properties and combinatorial structures of Subgroup graph $\Gamma(G)$ and obtain that the subgroup graph $\Gamma(G)$ of $G$ is never bipartite. Further, we show isomorphism and homomorphism of the Subgroup graphs of finite groups.
\end{abstract} 


\maketitle 
\section{Introduction}
\noindent One of the mathematical tools for studying symmetries of object is group theory, hence, several structures in the field of algebra are depicted through groups.  This mathematical concept has evolved rapidly since it discovery in the sixteenth century. According to \cite{Isra}, the rebirth of the axiomatic method and the view of mathematics as a human activity in the nineteenth century forms the major development that change the bearing on the evolution of group theory as a mathematical concept.   \cite{Isra}, further noted that the evolution had caused the previous classical algebra polynomial equations transited to the modern algebra of axiomatic systems of the nineteenth century. Meanwhile, this concept has been applied in the field of physics, chemistry and biology, (see \cite{Jian, Laan, Edwa,  Osbo}) for details.\\

\noindent In the same vein, in the last two decades, many studies have related graphs to group theory, providing a more easier way to visualize the concept of group; this relation brings together two important branches of mathematics, and has opened up a new wave of research with a better understanding of the fields. Many years after Euler's research work on the bridges of Königsberg city, Cayley \cite{Cal} used the generators of a finite group $G$ to define a graphical structure called the Cayley graph of finite group $G$, he further showed that every group of order $n$ can be represented by a strongly connected diagraph of $n$ vertices \cite{Kand}. Afterwards, in the last few decades, his view of diagraph has since been extended to different and modified graph of algebraic structures. Hence, more algebraic studies through the properties of these modified graphs have become topics of interest to many around the globe (See \cite{Paul, Cam, Reh, Wil, Xwa, Tolu, Akba, Andr}).

\noindent This study, the subgroup graph of finite groups $\Gamma(G)$ like \cite{Paul, Cam, Reh, Wil, Xwa, Tolu, Akba, Andr}, will focus on finite groups $G$, however, the choice of it vertex set $V(\Gamma(G))$, is the subgroups $S(G)$ of $G$.  In the literature, vertex set of graphs of finite groups are always the elements $n \in G$, a deviation from this norm is the motivation for this study. 

 
\subsection{Preliminaries}
\noindent We state some known and useful results which will be needed in the proof of our main results and understanding of this paper. For the definitions of the basic terms and results given in this section (see \cite {Gore, Robi, Dr, Gb, Anj}).\\

\noindent A \textit{graph} $\Gamma$ is a combinatorial structure formed by finite non-empty set (V, E), where $V$ is the set of vertices viewed as points and $E$ is the set of edges viewed as line joining the points. The cardinality of V($\Gamma$) is called the \textit{order} of $\Gamma$ while the cardinality of E($\Gamma$) is called the \textit{size} of $\Gamma$. The degree of a vertex $x$ in a graph $\Gamma$ denoted by $\delta(x)$ is the number of edges incident to it, that is the number of edges connecting $x$. A graph is said to have \textit{parallel edges} if there are more than one edges which join the same pair of distinct vertices. A \textit{loop} on the other hand is an edge that joins a vertex to itself while a \textit{walk} of length $k\leq n$ in a graph $\Gamma$ with vertex set $V(\Gamma)$ consist of an alternating sequence of vertices and edges consecutive elements of which are incident, that begins and ends with a vertex.

\begin{definition} \cite{Anj}
A \textit{complete graph} is a simple undirected graph in which has at least one vertex and every arbitrary pair of distinct vertices is joint by a unique edge. while a connected graph on the other hand is a graph where there is an edge between every pair of vertices.
\end{definition}
\begin{remark}
Note that every complete graph is necessary connected but connected graphs are not necessary complete. 
\end{remark}
\begin{definition} \cite{Anj}
A walk in a connected graph that visits every vertex of the graph exactly once without repeating the edges is called \textit{Hamiltonian path}. If this walk starts and ends at the same vertex, the walk is called a Hamiltonian circuit or cycle. 
\end{definition}
\begin{remark}
A graph that contains a \textit{Hamiltonian cycle} is said to be \textit{Hamiltonian}. 
\end{remark}
\begin{theorem}\label{thr0} (Sylow's First Theorem) \cite{Cam1}
Let $G$ be a finite group of order $p^rq$, where $p$ is a prime, $r$ and $q$ are positive integers and gcd(p,q) =1. Then $G$ has a subgroup of order $p^k$ for all $k$ satisfying $0\leq k \leq r$.
\end{theorem}
\begin{definition} \label{D4} \cite{Anj}
In graph theory, a \textit{regular graph} is a graph where each vertex has the same number of neighbors, i.e. every  vertex has the same degree or valency; a regular graph can be an x-regular graph where every vertex of the graph have the same degree $x$.
\end{definition}
\begin{definition} \cite{Anj}
The distance between two vertices $x,y \in V(\Gamma(G))$ is the \textit{length} of the shortest path between $x$ and $y$ and it's denoted by $\delta(x,y)$. \textit{Eccentricity} $\delta(x)$ of a vertex $x$ in a graph is define as $\delta(x)$ = max$\lbrace \delta(x,y):\forall x,y\in V(G) \rbrace$. The minimum and maximum eccentricity in a graph are called \textit{radius}, rad$(G)$ and \textit{diameter}, diam$(G)$ of the graph respectively.
\end{definition}
\begin{lemma} \cite{Anj} \label{Lem4}
Let $G$ and $G'$ be any two finite groups. If $G \cong G'$, then $S'(G) \cong S'(G')$ and $S(G) \cong S(G')$. But the converse is not true.
\end{lemma}

\begin{definition} \cite{Cam1}
A \textit{group} consists of a set $G$ with a binary operation * on $G$ satisfying the following
four conditions:
\begin{itemize}
\item[1.] Closure: $\forall a, b \in G$, we have $a*b \in G$.\\

\item[2.] Associativity: $\forall a, b, c \in G$, we have  $a(b*c)=(a*b)c$.\\

\item[3.] Identity: There is an element $e \in G$ satisfying $e*a = a*e = a$ for all $a \in G$.\\

\item[4.] Inverse: For all $a \in G$, there is an element $a^{-1}\in G$ satisfying $a*a^{-1} = a^{-1}*a = e$. (where $e$ is as in the Identity Law)\\
\end{itemize}

\end{definition}

\begin{definition} \cite{Anj}
A finite group is a group containing finite number of elements. The order of a finite group $G$ is the number of elements in $G$.
\end{definition}


\begin{definition} \cite{Cam1}
A subset $H$ of a group $G$ is called a subgroup if it forms a group in its own right with respect to the same operation on $G$.
\end{definition}

\begin{definition} \cite{Cam1}
Let $G_1$ and $G_2$ be groups. A homomorphism from $G_1$ to $G_2$ is a map $\theta$ which preserves the group operation.
\end{definition}

\begin{definition} \cite{Cam1}
A subgroup $H$ of $G$ is said to be a normal subgroup if it is the kernel of a homomorphism. Equivalently, $H$ is a normal subgroup if its left and right cosets coincide: $aH$ = $Ha$ for all $a \in G$. We write $"H$ is a normal subgroup of $G"$ as $H \unlhd G$; if $H \neq G$, we write $H \lhd G$.  If $H$ is a normal subgroup of $G$, we denote the set of (left or right) cosets by $G/H$. We define an operation on $G/H$ by the rule $(Ha)(Hb)$ = $Hab$ for all $a,b \in G$.
\end{definition}


\begin{definition} \cite{Woj}
Let $a$ and $b$ be elements of a group $G$ such that [a,b] yields an element of $G$ and is defined by $[a,b]=a^{-1}.b^{-1}.a.b$, the collections of arbitrary [a,b] in $G$ forms the commutator subgroup of $G$.
\end{definition}

\begin{lemma} \label{Lem} \cite{Gore}
Suppose $a$, $b$, $c$ $\in G$ and $e$ is a positive integer. Then
\begin{itemize}
\item[1.] $[a,b] = [b,a]^{-1}$.\\

\item[2.] $[a,bc] = [a,c][a,b]^c$.\\

\item[3.] $[ab,c] = [a,c]^b[b,c]$.\\

\item[4.] $[a,b^e] = \prod^{e-1}_{i=0}[a,b]^{b^i}$.\\
\end{itemize}

\end{lemma}

\begin{lemma} \label{Lem1} \cite{Robe}
Let $G$ be a group with $P$ a $p-$subgroup of $G$ and $T$ a $p'-$subgroup of $N_G(P)$. Set $R = [T, P]$, and the following holds
\begin{itemize}
\item[1.] $[T,R] = R,$ and if $P$ is abelian, then $P = R$ * $C_P(T)$.\\

\item[2.] If $R$ is abelian, then the minimal number of elements needed to generate $R$; $d(R) \leq 2$ and $t \in T'-C_\tau(P),$ then $[t,R] = R.$.\\

\item[3.] If $R$ is abelian and the minimal number of elements needed to generate $R$; $d(R) \leq 2$, there exists $t \in T$ such that $[t, R] = R$.\\

\item[4.]If $R$ is cyclic, then $T/C_\tau(P)$ is cyclic\\
\end{itemize}

\end{lemma}

\begin{lemma} \label{Lem2} \cite{Robe}
Suppose $G = \big<x,y, A \big>$ is a finite group with $A \geq G'$ abelian and $a = [x,y]$ of order $n$. If $C$ is the commutator subgroup of $G$, then $\{a^eb\mid b \in [G, A], (e, n) = 1\} \subseteq C$.
\end{lemma}

\begin{theorem} \label{Tm1} \cite{Robe}
Let $G$ be a $p-$group with $G'$ abelian and $d(G') \leq 2$. Set $C = C_G(G'/\Phi(G'))$.
\begin{itemize}
\item[1.] There exists $y \in G$ with $G' = [y, C]$. In particular, if $C < G$, then $y$ can be taken to be any element of $G - C$.\\

\item[2.] Let $[y, C] = G'$. If either $p \neq 2$ or $[G', C] \leq \Omega^2G'[y, G']$, then $G' = \Gamma_y(C)$.\\

\item[3.] $G'$ is equal to the commutator subgroup of $G$.\\

\end{itemize}

\end{theorem}

\begin{lemma} \label{Lem3} \cite{Robe}
Let $P$ be a $p-$subgroup of $G$, $t$ a $p'-$element of $N_G(P)$, and $s \in P$. If $[t, P]$ is abelian, then $\Gamma_{ts}(P) = [t, P]\Gamma_s(P)$.
\end{lemma}

\begin{theorem} \label{Th0} \cite{Peo}
Let $G$ be a group and let $H$, $K$ be two subgroups of $G$ and define $HK = \{hk : h \in H, k \in K\}$, then
\begin{itemize}
\item[1.] If both $H$ and $K$ are normal in $G$, then $H \cap K$ is also a normal subgroup of G.\\

\item[2.] If $H$ alone is normal in $G$, then $H \cap K$ is a normal subgroup of K.\\

\item[3.] If $H$ is normal in $G$, then $HK = KH$ and $HK$ is a normal subgroup of G.\\

\item[4.] If both $H$ and $K$ are normal in $G$, then $HK$ is a normal subgroup of G.\\

\end{itemize}
\end{theorem}

\begin{theorem}(Sylow's theorems) \label{Th011}
Let $G$ be a group of order $p^{\alpha}m$, where $p$ is a prime, $m \geq 1$, and $p$ does not divide $m$. Then:
\begin{itemize}
\item[1.] $Syl_p \neq \emptyset $, that is Sylow $p-$subgroups exist.\\

\item[2.] All Sylow p-subgroups are conjugate in $G$.\\

\item[3.] Any p-subgroup of $G$ is contained in a Sylow p-subgroup.\\

\item[4.] $n_p(G) \equiv 1$ mod p.\\

\end{itemize}
\end{theorem}


\begin{lemma} \label{Th1}
(Lagrange's theorem) \cite{Rose} Let $H$ be a subgroup of a finite group $G.$ Then the order of $H$ divides the order of $G$.
\end{lemma}

\section{Methods}
\noindent This article is not a variable base research, however, well known algebraic definitions and results were used to investigate the algebraic and combinatorial properties of the subgroup graph of finite groups.

\section{Results and Discussion}
\noindent The Subgroup graphs of finite groups is introduce in this section. We begin with the definition and notion of the Subgroups graph of a finite group.
\begin{definition} \label{D1}
Let $G$ be a finite group and $S(G)$ be the set of subgroups of $G$. Then the Subgroup graph of $G$ is the graph $\Gamma(G)$ with vertex set $V(\Gamma(G) = S(G)$ and the edge set
\[E(\Gamma(G)) = \{\{H, K\} : H \neq K, H \leq K or K \leq G\}.\]  
\end{definition}

\begin{remark}
Let $G$ be a finite group of order $n$, some clear consequences of the definition of subgroup graphs of finite groups $\Gamma(G)$ are
\begin{itemize} \label{Rm1}
\item[1.] The subgroup graph $\Gamma(G)$ is a simple graphs, thus, there are no loops nor multiple edges.\\

\item[2.]   The trivial subgroups of $G$ are adjacent to every other vertices on $\Gamma(G)$.\\

\item[3.]   Since the trivial subgroups of $G$ are adjacent to every other vertices on $\Gamma(G)$, then the graph is connected.\\

\item[4.]   The $\Gamma(G)$ has a diameter and radius of 1 if $\mid G \mid > 2$.
\end{itemize}
\end{remark}

\noindent Below, we give an example of Subgroups graph.
\begin{example}
Let $G$ = the group of integer modulo 6 under addition $(\mathbb{Z}_6,+)$, then the following is the undirected Subgroups graph of $(\mathbb{Z}_6)$.
\begin{center}

\usetikzlibrary{positioning} \begin{tikzpicture}[xscale=6, yscale=8,>=stealth] 
\tikzstyle{v}=[circle, minimum size=1mm,draw,thick] 
\node[v] (Z1) {$e$};
[right=of Z1]{};  \node[v] (Z2) [right=of Z1]{$\mathbb{Z}_2$}; 
[below=of Z1]{}; \node[v] (Z3) [below=of Z1]{$\mathbb{Z}_3$}; 
 [right=of Z3]{}; \node[v] (Z6) [right=of Z3]{$\mathbb{Z}_6$}; 
\draw[thick,-] (Z1) to node {}(Z2);
\draw[thick,-] (Z1) to node {}(Z3);
\draw[thick,-] (Z1) to node {}(Z6);
 \draw[thick,-] (Z2) to node {}(Z6);
 \draw[thick,-] (Z3) to node {}(Z6);
 \draw[thick,-] (Z2) to node {}(Z3);
\end{tikzpicture} \vspace{3mm}

\textit{Figure 1:} Subgroups graph of $\mathbb{Z}_6 $ \vspace{3mm}
\end{center}
\end{example}

\begin{theorem} \cite{Anj}
A simple graph is bipartite if and only if it does not have any odd cycle.
\end{theorem} 

\begin{remark}
Let $G$ be a finite group, the subgroup graph $\Gamma(G)$ of $G$ is never bipartite.
\end{remark}

\begin{theorem} \label{T1}
Let $G$ be a finite group of prime order and let $S(G)$ be the set of subgroups of $G$, then the subgroups graph of $G$ is a straight line with only two vertices.
\end{theorem}
\noindent \textbf{Proof:} Let $G$ be a finite group of prime order and let $S(G)$ be the set of subgroups of $G$, then by Lemma \ref{Th1}, the order of every $H \in S(G)$ divides the order of $G$ but the order of $G$ is a prime which can only be divided by itself of 1. Thus the only subgroups of this group $G$ are $H =\{ \{e\}$ and $G$ itself$\}$; and they are adjacent. \qed
\begin{remark}
Let $G$ be a finite group and let $\Gamma(G)$ be the subgroups graph of $G$. Then the vertex set $V(\Gamma(G)) \geq 2$ and edge set $E(\Gamma(G)) \neq \emptyset $; therefore, the subgroups graph of a finite group $G$ can never be empty.
\end{remark}

\begin{theorem}
Let $G$ be a group and let $H$, $K$ be two subgroups of $G$ and let $\Gamma(G)$ be the subgroup graph of $G$. Define $HK = \{hk : h \in H, k \in K\}$, then
\begin{itemize}
\item[1.] If both $H$ and $K$ are normal in $G$, then $H \cap K$ is also a vertex on $\Gamma(G)$.\\

\item[2.] If $H$ alone is normal in $G$, then $H \cap K$ is a adjacent to vertex $K$ on $\Gamma(G)$.\\

\item[3.] If $H$ is normal in $G$, then $HK = KH$ and $HK$ is also a vertex on $\Gamma(G)$.\\

\item[4.] If both $H$ and $K$ are normal in $G$, then $HK$ is also a vetex on $\Gamma(G)$\\

\end{itemize}
\end{theorem}

\noindent \textbf{Proof:} From Theorem \ref{Th0}, the results follows.\qed

\begin{theorem}
Let $G$ be a finite group of non prime order $n > 2$, then the subgroup graph of G $(\Gamma(G))$ is never a star graph
\end{theorem}
\noindent \textbf{Proof:} Suppose on the contrary, let the subgroup graph of a finite group of non prime order $n > 2$ be a star graph; then it implies that all other $H \in S(G)$ are subgroup to only an arbitrary subgroup $K \subseteq G$, but by Remark \ref{Rm1}(2), every group has two trivial subgroups which are adjacent to all other $H \in S(G)$. So, the graph can not be a star graph, since, there is more than one vertex that is adjacent to all the vertices of $\Gamma(G)$. \qed

\begin{theorem}
Let $G$ and $G'$ be two isomorphic finite groups. Then the subgroup graph of $G$ is isomorphic to subgroup graph of $G'$ $(\Gamma(G) \cong \Gamma(G'))$.
\end{theorem}

\noindent \textbf{Proof:} Suppose $G$ and $G'$ are two isomorphic finite groups, then from Lemma \ref{Lem4}, $S(G) \cong S(G')$. Thus, $(\Gamma(G) \cong \Gamma(G'))$. \qed

\begin{theorem}
Let $G$ be a group of order $p^{\alpha}m$, where $p$ is a prime, $m \geq 1$ and $p$ does not divide $m$. Suppose $\Gamma(G)$ is the subgroup graph of $G$, $V(\Gamma(G))$ and $E(\Gamma(G))$ are the vertex and edge sets of $\Gamma(G)$ respectively. Then:
\begin{itemize}
\item[1.] There must exist Sylow p-subgroups of $G$; $H, K \in V(\Gamma(G))$ which are conjugate in $G$ and are adjacent to each other on the subgroup graph of G.\\

\item[2.] There is a vertex $H$ on the subgroup graph of $G$ that is adjacent to a vertex $K$ that is a Sylow p-subgroup of G.

\end{itemize}
\end{theorem}

\noindent \textbf{Proof:}
\begin{itemize}
\item[1.] To show that there must exist Sylow p-subgroups of $G$; $H, K \in V(\Gamma(G))$ which are conjugate in $G$ and are adjacent to each other on the subgroup graph of G, then it suffices if we can show $H, K \in V(\Gamma(G))$ to be conjugate and subsequently establish an edge between $H$ and $K$. From Theorem \ref{Th011}, it is established that the group $G$ has some Sylow p-subgroups. So, let $H$ be a Sylow p-subgroup of G, and let $S$ be the set of all distinct conjugates of $H$. Suppose $k$ is the order $S$, we need to establish that $p$ cannot divide $k =\  \mid S \mid $.  Since each of the $K \in S$ is a conjugate to $H$, it implies every element of $S$ is in the orbit of $H$. So using the formula for orbit size $\mid S \mid = $ $\mid$ orbit of $H \mid = (G:N_G(H))$ (where $N_G(H))$ is the normalizer of $H$). However, Lagrange's theorem established that $\mid G \mid = (G:N_G(H))\mid$ and clearly, $H$ is a subgroup of $N_G(H)$ and it contains $p^\alpha$ as a factor and a maximum power $p$ can assume. Thus, $(\mid G \mid $ divides $\mid N_G(H)\mid$ and $\mid S \mid$ contains no factor of $p$ and so $p$ does not divide $\mid S \mid$.\\

\item[2.] Suppose $H$ is any $p$-subgroup of $G$, it will suffice if we can show $H \leq K_i$ $i = 1, ..., p$. Let $H$ act on $K = \{K_i ... K_p\}$, by conjugation. Clearly, the the orbits of this action will partition $K$. Suppose the distinct orbits are the orbits of $\{K_{1i} ... K_{1p}\}$ then the orbit of $\mid K \mid = \mid orbit K_{1i} \mid + ... + \mid orbit K_{1p} \mid \}$. To compute the orbit for any $K_i \in K$, $\mid$ orbit of $ K_i \mid = (H : H \cap N_G(K_i)$, since $H \cap N_G(K_i)$ is the stabilizer of $K_i$ under the action of $H$. Then the size of each orbit has to divide $\mid H \mid$, which is a power of $p$. Though $p$ doesn't divide $\mid H \mid $ so there is no $p$ dividing all the terms $\mid$ orbit of $P_{1i}\mid$, ..., $\mid$ orbit of $P_{1i}\mid$ and else $p$ would divide their sum and also $K$. Assume 1 = $\mid$ orbit of $ K_{1i} \mid = (H : H \cap N_G(K_i)$  which means $(H = H \cap N_G(K_i)$ and since $H$ is a $p-$subgroup $H \cap N_G(K_i) = H \cap K_i$. This implies $H = H \cap K_i$ so every element of $H$ is also in $K_i$ then $H \leq K$, therefore, the $p-$subgroup $H$ is a subgroup of the arbitrary Sylow $p-$subgroup of $G$. So, $H$ and $K$ are adjacent of the subgroup graph $\Gamma(G)$ of $G$. \qed

\end{itemize}

\begin{theorem}
Let $G$ and $G'$ be two finite groups and $\phi : G \rightarrow G'$ be a group homomorphism. Suppose there is an $N$, a normal subgroup of $G$ and an $N'$, a normal subgroup of $G'$ such that $N$ is adjacent to $G$ on $\Gamma(G)$ the subgroup graph of $G$  and $N'$ is adjacent to $G'$ on $\Gamma(G')$ the subgroup graph of $G'$. Then
\begin{itemize}
\item[1.] $\phi(N)$ is adjacent to $\phi(G)$ on $\Gamma(\phi(G))$ the subgroup graph of $\phi(G)$.\\

\item[2.] $\phi^{-1}(N)$ is adjacent to $G$ on $\Gamma(G)$ the subgroup graph of $G$.\\
\end{itemize}
\end{theorem}

\noindent \textbf{Proof:} Suppose $G$ and $G'$ are two finite groups and $\phi : G \rightarrow G'$ is a group homomorphism, if there is an $N$, a normal subgroup of $G$ and an $N'$, a normal subgroup of $G'$ such that $N$ is adjacent to $G$ on $\Gamma(G)$ and $N'$ is adjacent to $G'$ on $\Gamma(G')$ then to show that $\phi(N)$ is also adjacent to $\phi(G)$ on $\Gamma(\phi(G))$ and  $\phi^{-1}(N)$ is adjacent to $G$ on $\Gamma(G)$; it will suffice if we can show $\phi(N)$ to be a normal subgroup of $\phi(G)$ and $\phi^{-1}(N)$ to be a normal subgroup of $G$ respectively. 
\begin{itemize}
\item[1.] Let $\phi(g) \in \phi(G)$, since $\phi$ is a group homomorphism and $N$ is normal in G.\
$\phi(g)\phi(N)\phi(g)^{-1}$ = $\phi(gNg^{-1})$ = $\phi(N)$. Thus, $\phi(N)$ is normal in $\phi(G)$.\\

\item[2.] Let $a$ be an arbitrary element of $G$, then the set $a\phi^{-1}(N')a^{-1}$ satisfies that\\ $\phi(a\phi^{-1}(N')a^{-1})$ = $\phi(a)\phi(\phi^{-1}(N')\phi(a^{-1}) \subseteq \phi(a)N'\phi(a)^{-1} \subseteq N'$ since $N'$ is normal in $G'$. Thus, $a\phi^{-1}(N')a \subseteq \phi^{-1}(N')$ for every $a \in G$. This shows that $\phi^{-1}(N')$ is a normal subgroup of $G$.\\ \qed
\end{itemize}

\begin{corollary} \cite{Note, Keit5} \label{Tt1}
The alternating group $A_n$ is a subgroup of the symmetric group $S_n$.
\end{corollary}

\begin{theorem}
Let $\Gamma(S_n), \Gamma(D_n)$ and $\Gamma(A_n)$ be the subgroups graphs of symmetric groups, $S_n$, dihedral groups $D_n$ and the alternating groups $A_n$, $n\geq 3$. Suppose $K$ and $M$ are vertices on $\Gamma(D_n)$ and $\Gamma(A_n)$ respectively, then both $K$ and $M$ are also vertices on $\Gamma(S_n)$ and are adjacent to $S_n$ if and only if $K$ is adjacent to $D_n$ on $\Gamma(D_n)$ and $M$ is adjacent to $A_n$ on $\Gamma(A_n)$.
\end{theorem}
\noindent \textbf{Proof:} Let $\Gamma(S_n), \Gamma(D_n)$ and $\Gamma(A_n)$ be the subgroups graphs of the symmetric groups, $S_n$, dihedral groups $D_n$ and the alternating groups $A_n$, $n\geq 3$. To show that $K$ and $M$ which are vertices on $\Gamma(D_n)$ and $\Gamma(A_n)$ respectively are also vertices on $\Gamma(S_n)$ and are adjacent to $S_n$ if and only if $K$ is adjacent to $D_n$ on $\Gamma(D_n)$ and $M$ is adjacent to $A_n$ on $\Gamma(A_n)$, then it suffices, if we can show both $K$ and $M$ to be subgroups of $S_n$.\\
Assume that $K$ is a subgroup of $D_n$ and $M$ is a subgroup of $A_n$. Then observe the structure of the group of symmetries of a regular $n-$gon in a plane (dihedral group ($D_n$)), it is isomorphic to a subgroup of $S_n$ then it is a proper subgroup of $S_n$. Also, by Corollary \ref{Tt1}, $A_n$ is a subgroup of $S_n$. Moreover, since $K \leq D_n$ and $M \leq A_n$ by implication, they are also subgroups of $S_n$ and hence are vertices on $\Gamma(S_n)$.\\
Conversely, assume that $K$ is adjacent to $D_n$ on $\Gamma(D_n)$ and $M$ is adjacent to $A_n$ on $\Gamma(A_n)$, then by Definition \ref{D1}, $K \leq D_n$ and $M \leq A_n$. But both $D_n$ and $A_n$ are subgroups of $S_n$, also, by implication, they are adjacent to $S_n$. \qed

\begin{theorem} \cite{Step}
If $n\geq 3$, then the number of subgroups of the dihederal group $D_n$ is $\tau(n) + \sigma(n)$. Where $\tau(n)$ is the number of divisors of $n$ and $\sigma(n)$ is the sum of divisors of $n$.
\end{theorem}

\begin{remark}
Let $D_n$ be a dihedral group of order $n\geq 3$, then the number of vertices on the subgroups graphs $\Gamma(D_n)$ of the dihederal group $D_n$ is $\tau(n) + \sigma(n)$, where $\tau(n)$ is the number of divisors of $n$ and $\sigma(n)$ is the sum of divisors of $n$.
\end{remark}

\begin{theorem}
Let $C$ be a commutator subgroup of a finite group $G$ of order $n$, suppose there exist a normal subgroup $N$ of $G$ such that $G/N$ is Abelian, then $C$ and $N$ are adjacent on the subgroup graphs of G.
\end{theorem}

\noindent \textbf{Proof:} Let $G$ be a finite group of order $n$ and $C$ the commutator subgroup of $G$. If there exist a normal subgroup $N$ of $G$ such that $G/N$ is Abelian, then to show that $C$ and $N$ are adjacent on $\Gamma(G)$ (the subgroup graph of $G$), we must show that either $C \leq N$ or $N \leq C$. Note that $N$ is normal in $G$ and $G/N$ is Abelian, then for $x,y \in G$, we have $(xN)(yN)$ = $(yN)(xN)$ and using the definition of Coset multiplication $xyN = yxN$. Which implies $xy(yx)^{-1} \in N$, where $xy(yx)^{-1} = xyx^{-1}y^{-1}$. Similarly, $xyx^{-1}y^{-1} \in N$ and since $x$ and $y$ are arbitrary then any commutator in $G$ is an element of $N$ and since $N$ is a subgroup of $G$ then any finite product of commutators in $G$ is an element of $N$ and thus $C \leq N$. \qed

\begin{theorem}
Let $C$ be a commutator subgroup of a finite group $G$ of order $n$ and let $H$ be a subgroup of $G$. If there exist an $x \in G$ such that $[x,H] \subseteq C_G(H)$, then $[x,H]$ and $C$ are adjacent on the subgroup graphs of G.
\end{theorem}

\noindent \textbf{Proof:} By Lemma \ref{Lem}(2) and using the method of \cite{Robe}, the map sending $h \in H$ to $[x,h^{-1}]$ is a homomorphism.Thus, the image map of $H$ is the subgroup $[x, H]$.  \qed

\begin{theorem} (Schur Zassenhaus) \label{SZ} \cite{Gore}
Let $G$ be a finite group and write $\mid G \mid = ab$ where $(a,b) = 1$. If $G$ has a normal subgroup of order $a$ then it has a subgroup of order $b$.
\end{theorem}

\begin{remark}
Let $G$ be a finite group and write $\mid G \mid = ab$ where $(a,b) = 1$. Then the vertex set $V(\Gamma(G))$ of the subgroup graph of $G$ contains at least two vertices $H, K \leq G$ which orders are $a$ and $b$ respectively.
\end{remark}

\begin{theorem}
Let $G$ be a finite nilpotent group, such that $G'$ is an abelian $p-$group with the minimal number of elements needed to generate $G'$; $d(G') \leq 3$, then $G'$ is a vertex on the subgroup graph $\Gamma(G)$ and it is adjacent to $G$.
\end{theorem}

\noindent \textbf{Proof:} Suppose $G$ is a finite nilpotent group, such that $G'$ is an abelian $p-$group with $d(G') \leq 3$, then to show that $G'$ is a vertex adjacent to $G$ on the subgroup graph $\Gamma(G)$ of $G$, it will suffice if we can show $G'$ to be equal to $C$, the commutator subgroup of $G$. Now, since $G$ is finite, we assume an arbitrary $P \in Syl_p(G)$ such that $G' \leq P$ and obviously, $P$ is normal in $G$ and by Theorem \ref{SZ} (Schur Zassenhaus theorem) and the methods in \cite{Robe}, $G = PQ$ where $P \cap Q = \{e\}$. Also, by Lemma \ref{Lem1}(3), and Lemma \ref{Lem2}, we set $R = [T, P]$ and if $G' = R$ there exist $t_1, t_2 \in T$ such that $R = [t_1, R][t_2, R]$ and $G' = \{[t_1a, t_2b]\mid a, b \in R\} = C.$ Thus, we can assume $d(R)\leq 2$, and they also exists $t \in T$ with $R = [t, P]$. Since $G/R$ is nilpotent, by Theorem \ref{Tm1} and Lemma \ref{Lem3},
\[G' = R(C) = R \big(\bigcup_{s \in P}\Gamma_s(P)\big)= \bigcup_{s\in P}\Gamma_{ts}(P) = C\] \qed

\begin{lemma}
Let $G$ and $H$ be two non nilpotent finite groups, such that there is an isomorphism $\phi$ of $G'$ and $H'$, if the commutator subgroup $C$ of $G$ is adjacent to $G$ on the subgroups graph $\Gamma(G)$ of $G$, then the commutator subgroup $C$ of $H$ is also adjacent to $H$ on the subgroups graph $\Gamma(H)$ of $H$.
\end{lemma}

\noindent \textbf{Proof:} Suppose $G$ and $H$ are non nilpotent finite groups such that there is an isomorphic map between $G'$ and $H'$, then we can safely say there is also isomorphic map between the commutator subgroups of $G'$ and $H'$, which shows the isomorphic relationship between $G$ and $H$. Also, since the commutator subgroup of $G$ is adjacent to $G$ on $\Gamma(G)$ then the commutator subgroup of $H$ is also adjacent to $H$ on $\Gamma(H)$.  \qed


\begin{example}
Let $Q_8$ be a quaternion group generated by the following matrices 
\[ 1 = \left( \begin{array}{cc}
1 & 0 \\
0 & 1
\end{array} \right),
%
i = \left( \begin{array}{cc}
i & 0 \\
0 & -i
\end{array} \right),
%
j = \left( \begin{array}{cc}
0 & 1 \\
-1 & 0
\end{array} \right),
%
k = \left( \begin{array}{cc}
0 & i \\
i & 0
\end{array} \right).
%)
\]\cite{Mari}, using the matrix multiplication obtained $Q_8$ = $\{\pm 1, \pm i, \pm j, \pm k\}$, observe that the subgroups of $Q_8$ consists of $Q_8$ itself and of the cyclic subgroups $\big<1\big> = \{e\}$, $\big<-1\big> = \{1,-1\}$ ,$\big<i\big> = \{1,i,-1,-i\}$,$\big<j\big> = \{1,j,-1,-j\}$,$\big<k\big> = \{1,k,-1,-k\}$ and the following is the subgroups graph of $Q_8$.
\begin{center}

\usetikzlibrary{positioning} \begin{tikzpicture}[xscale=6, yscale=8,>=stealth] 
\tikzstyle{v}=[circle, minimum size=1mm,draw,thick] 
\node[v] (Z1) {$e$};
[right=of Z1]{};  \node[v] (Z2) [right=of Z1]{$-1$}; 
[above=of Z2]{};  \node[v] (Z5) [above=of Z2]{$\mathbb{Q}_8$}; 
[left=of Z5]{}; \node[v] (Z3) [left=of Z5]{$k$}; 
[right=of Z5]{}; \node[v] (Z6) [right=of Z5]{$i$}; 
[below=of Z6]{}; \node[v] (Z4) [below=of Z6]{$j$}; 
\draw[thick,-] (Z1) to node {}(Z2);
\draw[thick,-] (Z1) to node {}(Z3);
\draw[thick,-] (Z1) to node {}(Z4);
 \draw[thick,-] (Z1) to node {}(Z5);
 \draw[thick,-] (Z1) to node {}(Z6);
  \draw[thick,-] (Z2) to node {}(Z3);
  \draw[thick,-] (Z2) to node {}(Z4);
  \draw[thick,-] (Z2) to node {}(Z5);
  \draw[thick,-] (Z2) to node {}(Z6);
 \draw[thick,-] (Z3) to node {}(Z5);
  \draw[thick,-] (Z6) to node {}(Z5);
  \draw[thick,-] (Z4) to node {}(Z5);
 \end{tikzpicture} \vspace{3mm}

\textit{Figure 2:} Subgroups graph of $\mathbb{Q}_8 $ \vspace{3mm}
\end{center}
\end{example}

\section{Conclusions}
This study has highlighted some algebraic properties and combinatorial structures of Subgroups graph $\Gamma(G)$ of finite groups. The connections between the Subgroups graphs of finite groups upto homomorphism and isomorphism were also studied and further looked at the relationships between the subgroups graphs of symmetric groups, $S_n$, dihedral groups $D_n$ and the alternating groups $A_n$, when $n\geq 3$.
%
\section{Abbreviations}
Not applicable
\section*{Declarations}
\subsection{Availability of data and materials}
\noindent Data sharing not applicable to this article as no data sets were generated or analysed during the current study.
\subsection{Competing interests}
The authors declare that they have no competing interests.
\subsection{Funding}
There are no funding for this research work.
\subsection{Authors' contribution}
OE conceived the study, wrote the results and proofs, AIG validated the results and help in proofreading the article while KOA reviewed the literature and preliminaries. All authors have read and approved the final manuscript.
\subsection{Acknowledgments}
The authors wish to thank the reviewers for their helpful comments and recommendations on this article.


 


  









\end{document